Author’s response to reviews

Title: Tongue metastasis as an initial presentation of renal cell carcinoma. A case report/series and review of indexed literature Corresponding author: Faisal Azam MRCPI, MRCP (UK), Staff grade in Oncology. Department of Oncology, North Wales Cancer Treatment Centre, Bodelwyddan, Rhyl, LL185UJ, UK Author 2: Muneer Abubakerr, SHO in Oncology, North Wales Cancer Treatment Centre, Bodelwyddan, Rhyl, LL185UJ, UK Author 3: Simon Gollins MRCP, FRCR, PhD, Consultant clinical Oncologist, Department of Oncology, North Wales Cancer Treatment Centre, Bodelwyddan, Rhyl, LL185UJ, UK

Authors:

Faisal Dr Azam MRCP (drfaisalazam@hotmail.co.uk)
Muneer Dr Abubakar (muneer_dr2002@yahoo.co.uk)
Simon Dr Gollins MRCP,FRCR,PhD (simon.gollins@cd-tr.wales.nhs.uk)

Version: 2 Date: 26 October 2007

Author’s response to reviews:

Dear Sir/Madam

Please consider the revised version of the case report for publication in your journal.

Thanks